Author's response to reviews

Title: Epigenetic Regulation and Role of Metastasis Suppressor Genes in Pancreatic Ductal Adenocarcinoma

Authors:

Wolf A Mardin (wolf.mardin@ukmuenster.de)
Joerg Haier (haier@uni-muenster.de)
Soeren T Mees (soerentorge.mees@ukmuenster.de)

Version: 2 Date: 8 April 2013

Author's response to reviews:

The revisions are provided in full in the cover letter which includes a point by point reply to the Editorial Board's comments. Further, the revisions have been highlighted in the revised manuscript.